36-303: Surveys, Sampling and Society Choosing Office Hours

1. Professor Junker is always available right after class to answer questions. He is also available by appointment (brian@stat.cmu.edu; 412-268-8874).

2. Jionglin Wu is the TA for this class. Please put a checkmark below next to each office hour you could make at least 1/2-hr of if you have questions about homework etc. (Note that calculation-style homework will typically be due on Thursdays.)

Mo 9-10am Mo 10-11am Mo 1:30-2:30pm Mo 2:30-3:30pm Tu 1-2pm Tu 1:30-2:30pm We 9-10am

We 10-11am Th 1-2pm Th 1:30-2:30pm Fr 9-10am Fr 10-11am Fr 3-4pm Fr 4-5pm

Note: All of Jionglin’s office hours will be held in the Statistics TA Offices in the Facilities Management Services (FMS) Building.

I will announce office hours later this week after compiling your responses. 3. If you have any other comments about office hours, or about class generally, please write them below.